Author's response to reviews

Title: Effect of Tualang honey on the healing of anastomotic wound in large bowel anastomosis in rats-A randomized controlled trial

Authors:

Muhammad Izani Aznan (alnerazzuri@yahoo.com)
Omaid Hayat Khan (omaidhayatkhan@hotmail.com)
Allah Obhayo Unar (unarobhayo@yahoo.com)
Sharifah Emilia Tuan Sharif (dremilia@usm.my)
Amer Hayat Khan (amerhayat@ymail.com)
Syed Hassan (shassan@usm.my)
Andee Dzulkarnain Zakaria (andee778@gmail.com)

Version: 4 Date: 31 December 2015

Author's response to reviews: see over
COVER LETTER

Dr. Andee Dzulkarnain Zakaria
Lecturer, Department of Surgery,
School of Medical Sciences,
Universiti Sains Malaysia-Health Campus
16150, Kota Bharu, Kelantan, Malaysia.

Dated: 30/12/15

Dear Editor BMC (C & AM),

I wish to submit a revised manuscript entitled “Effect of Tualang honey on the healing of anastomotic wound in large bowel anastomosis in rats - A randomized controlled trial” for consideration by BMC Complementary & Alternative Medicine journal.

With the submission of this revised manuscript I would like to address the questions from the reviewers as follows:

Reviewer: Scott Sell

As per the reviewers report, I see no further question except for a suggesting statement (if I may not be wrong) about the expansion of data. Due to our budgetary limitation we already explained we will consider more comprehensive study including all the suggested factors in our future studies. It has been already mentioned in our limitations section in previously edited version of the manuscript. Language correction has been double checked as well. Kindly highlight the concerns if any.
Reviewer: Jaroslava Halper

Thank you for appreciating and taking interest in our manuscript. However, as per your enquiry about the reference “Yol S et al.”, it was just a reference for the reviewer as we were asked to provide the reference from the literature to substantiate the duration of study and project design. This reference is not necessarily needed as to be the part of the manuscript. Language corrections have also been double checked.

Please address all correspondence concerning this manuscript to me at andee778@gmail.com

Thank you for your consideration of this manuscript.

Sincerely,

Andee Dzulkarnain Zakaria
Department of Surgery,
School of Medical Sciences,
Universiti Sains Malaysia, Health Campus
16150, Kota Bharu, Kelantan, Malaysia.
Tel: 006-012-9541936;
Fax: +6097648277
Email: andee778@gmail.com